Author’s response to reviews

Title: The effect of regular walks on various health aspects in older people with dementia: protocol of a randomized-controlled trial

Authors:

Karin M Volkers (k.m.volkers@vu.nl)
Erik J.A Scherder (eja.scherder@vu.nl)

Version: 3 Date: 8 August 2011

Author’s response to reviews:

I changed the order of mentioning the outcome domains in such a way that I always mention it in the same order as I do in the introduction. So I changed the order in the abstract (background) and in the method (procedure and assessments). The numbering of references mentioned in the assessments are correctly adjusted.

Further changes in the method:
study design: I changed 'and' into 'or' in the last sentence.
procedure: I added 'and pain' in the second sentence.
characteristics: I changed 'Dementia severity' into 'Stage of dementia', because I named it in the procedure also stage of dementia.
assessment of physical functioning: Figure of eight: I changed 'Figure of Eight' into 'figure of eight' in the second sentence and in the last sentence I changed '(sec)' into 'in seconds'.
assessment of physical functioning: Frailty and injuries: I changed 'FicSit' and 'Ficsit' into 'FICSIT'.
assessment of cognition: I changed 'Mini-mental state examination' into 'Mini-Mental State Examination'.
assessment of cognition: picture recognition: I added that it measures 'visual, verbal long term memory' instead of only 'verbal long term memory'
assessment of cognition: stroop task: I deleted 'at' in the last sentence.
assessment of rest-activity rhythm: IV: I changed 'normal' into 'better' in the last sentence.
assessment of rest-activity rhythm: RA: I added 'the relative difference' instead of 'the difference'.
assessment of QoL: I added 'by the nursing staff' instead of only 'by nursing staff'.
assessment of ADL: I added 'by the nursing staff' instead of only 'by nursing staff'.

In references:
One reference that was 'in press' is now published and therefore I added the publication information(ref 37).